



Article On Supporting University Communities in Indoor Wayfinding: An Inclusive Design Approach

Catia Prandi ^{1,2}, Giovanni Delnevo ^{1,*}, Paola Salomoni ¹ and Silvia Mirri ¹

- Department of Computer Science and Engineering, University of Bologna, 40126 Bologna, Italy;
- catia.prandi2@unibo.it (C.P.); paola.salomoni@unibo.it (P.S.); silvia.mirri@unibo.it (S.M.)
- ² ITI-LARSyS, 9020-105 Funchal, Portugal
 - Correspondence: giovanni.delnevo2@unibo.it

Abstract: Mobility can be defined as the ability of people to move, live and interact with the space. In this context, indoor mobility, in terms of indoor localization and wayfinding, is a relevant topic due to the challenges it presents, in comparison with outdoor mobility, where GPS is hardly exploited. Knowing how to move in an indoor environment can be crucial for people with disabilities, and in particular for blind users, but it can provide several advantages also to any person who is moving in an unfamiliar place. Following this line of thought, we employed an inclusive by design approach to implement and deploy a system that comprises an Internet of Things infrastructure and an accessible mobile application to provide wayfinding functions, targeting the University community. As a real word case study, we considered the University of Bologna, designing a system able to be deployed in buildings with different configurations and settings, considering also historical buildings. The final system has been evaluated in three different scenarios, considering three different target audiences (18 users in total): i. students with disabilities (i.e., visual and mobility impairments); ii. campus students; and iii. visitors and tourists. Results reveal that all the participants enjoyed the provided functions and the indoor localization strategy was fine enough to provide a good wayfinding experience.

Keywords: indoor wayfinding; indoor localization technologies; inclusive design; BLE beacon; Internet of Things; people with disabilities

1. Introduction

Finding the appropriate position and direction in the environment is an important issue for human beings since a long time ago. In fact, many of the most important human activities are based on the capabilities of orientating and independently moving in the space and in the surrounding environment [1,2].

Its importance is evident even in Greek mythology: Ariadne's string had a significant role, being a mean that let Theseus retrace his way out of the labyrinth of the Minotaur. Nowadays, modern Ariadne's strings are supporting people in orientating and in moving within buildings and across urban environments. Basically, they are the result of the exploitation of mobile devices equipped with sensors that receive signals from external sources, employing specific positioning and wayfinding algorithms.

Even though radio waves of the satellites propagate for enormous distances, in the proximity of the user they can encounter obstacles, such as walls, roofs, or vegetation that can inhibit their reception. For this reason, inside buildings, it is unlikely to receive a usable GPS signal to translate the navigation services designed for the outside into enclosed spaces. Therefore, it is necessary, in order to accurately localize individuals in indoor environment and to provide them wayfinding and navigation support, to find alternative technologies [3].

The wide diffusion of smartphones and the emergence of IoT infrastructures and smart objects have been only a small part of the huge technological revolution, that has



Citation: Prandi, C.; Delnevo, G.; Salomoni, P.; Mirri, S. On Supporting University Communities in Indoor Wayfinding: An Inclusive Design Approach. *Sensors* **2021**, *21*, 3134. https://doi.org/10.3390/s21093134

Academic Editor: Joanna Kolodziej

Received: 24 March 2021 Accepted: 28 April 2021 Published: 30 April 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). distinguished the last years and has seen an ever-growing interconnection and interaction between people and mobile devices. The paradigm shift envisioned by the Internet of Things represents a unique opportunity in supporting people, especially those with disabilities, who could get a significant benefit, improving their independence in conducting their daily life activities [4]. For instance, these activities are ranging from letting people with visual and/or mobility impairments control smart home lights, heating, and appliances in an easier way, to managing home security systems; from moving within a complex and unknown building, to get additional information about architectural elements and points of interest, and so on. An IoT focused on users' and communities' needs could really play a key role in terms of inclusiveness and accessibility [5].

In this context, taking into account indoor positioning and wayfinding, some questions arise: which are the most suited technologies for user's positioning in a certain context? Which design approach better lets to meet users and communities' needs?

In this paper, we aim to answer these questions, by presenting a case study based on the design, development, and evaluation of a pervasive system, named AlmaWhere. The main goal of this work is to study solutions so as to identify how to equip users with a system to navigate an indoor environment, localizing Points of Interest (PoIs). Specific attention has been paid in the design phase, with the aim of providing an inclusive tool, meeting the specific needs of users with disabilities, in particular those people with mobility impairments (who need tailored paths within a building, avoiding architectural barriers, such as stairs and steps) and people with visual impairments (who need specific landmarks, turn by turn instructions, verbose descriptions of the environment, as well as accessible user interfaces and interaction mechanisms). As per other accessibility techniques and solutions, we aimed to get benefits for a larger users' community, taking advantage of a curb-cut effect [6]. Indoor positioning and wayfinding have been based on the use of BLE (Bluetooth Low Energy) beacon devices, evaluating specific techniques by involving target users. AlmaWhere has been applied as a real case study at the University of Bologna (which is composed of five campuses in five different cities), this allowed us to conduct different field studies, by involving three communities, including students with disabilities and tourists who are interested in visiting the historical building of the most ancient university of the western world. This paper discusses the adopted techniques and design approach and it reports the results we obtained from the field study evaluations, carried out in three different scenarios and with three target groups.

The remainder of this paper is organized as follows. Section 2 presents the main related work, focusing on indoor positioning techniques and on main strategies to support users with disabilities in navigating and wayfinding activities. Section 3 describes the design issues that have driven our work. Section 4 illustrates the system walkthrough, presenting our real case scenario, while Section 5 reports the results of the evaluations we have conducted with target users. Finally, Section 6 concludes the paper by hinting at some future work.

2. Related Work

This Section is organized into two parts. In the first one, we present the state-of-the-art of indoor positioning, illustrating the different approaches that have been proposed over the years. The second one, instead, describes some projects devoted to the development of indoor wayfinding/navigation systems for people with disabilities, focusing on similarities and differences with the AlmaWhere project.

2.1. Indoor Positioning

The key element of any wayfinding system is its positioning system. While in outdoor environments the Global Positioning System (GPS) is an accurate and standard solution [7], the indoor localization is still an open challenge. Different approaches have been proposed to address this challenge. A first categorization can be done based on whether or not they require additional infrastructures. Being able to locate users without the need for any particular infrastructure is certainly an attractive prospect, given the obvious advantages regarding the costs (of installation and maintenance) and the scalability of the whole system. The first family of approaches, that falls in this group, is the one based on the real-time processing of images captured by smartphones camera. Essentially, computer vision techniques or deep learning algorithms are employed to detect specific markers in the environment, that allow us to understand the user position [8,9]. In the context of visually impaired users, the idea at the basis of such approaches is to replace the lost sense (i.e., the sight) with an artificial eye [10]. These approaches have the additional advantage of being able to detect obstacles [11]. However, similar approaches also present some major drawbacks. The most relevant is the possible

obstruction of the visual field by people and objects. In fact, for the correct functioning of this type of system, it is essential not to have obstacles between the camera and the markers, a constraint hard to satisfy in crowded environments. Another drawback concerns the high battery consumption resulting from the continuous use of the camera and from the real-time processing of the captured images. A variant of the latter approach utilizes visible light communication technology [12,13].

The other localization approach that does not require additional infrastructures is the one based on Wi-Fi signals. It falls in this category since it is possible to assume that the public (and private) buildings are equipped with Wi-Fi. This approach typically exploits the Received Signal Strength Indicator (RSSI) values from multiple access points to determine the users' position [14,15]. As can be easily deduced, the accuracy of such an approach strongly depends on the number, and the relative position, of the access points. This is an important limitation since the access points deployment is usually performed to guarantee an adequate internet connection and not to ensure precise localization in every place of the building. Moreover, since Wi-Fi signals are unstable and noisy, the received RSSI values in the same place could vary significantly over time, making localization difficult. To partially mitigate these issues, further access points may be added at specific points with the aim of improving the localization accuracy. Another option is to employ sensor fusion techniques to combine data coming from other smartphone sensors to RSSI [16].

With regard to the approaches that require additional infrastructure, the main ones are based on Radio-Frequency Identification (RFID) and beacons. RFID uses radio frequencies to perform wireless communication between a tag and a receiver. RFID tags can be active or passive [17,18]. Passive tags do not carry batteries but are powered by the reader signal. For this reason, the communication range is limited. On the other hand, active tags (equipped with batteries) can transmit at higher power levels and thus have a longer range. Despite this, the interest of the scientific community is primarily on passive tags. In fact, while active RFIDs are just another example of local-area communication technologies, passive tags present unique advantages from a system maintenance perspective: the limited cost per tag and no need for replacement due to low battery. Notwithstanding these benefits, the use of RFID in navigation systems remained limited due to the following factors [19]. First of all, the reader devices must be at a very short distance to be able to detect the passive tag, a clear issue for people with visual impairments [20]. Furthermore, RFID technology requires a specific receiver that is rarely included in smartphones or other commonly used objects [21].

Beacons, instead, are devices that use BLE to broadcast some data at a fixed time interval. They have emerged as one of the most promising technology in the Internet of Things (IoT) ecosystem [22]. Their power supply depends on the constructive choices of the producers. The most adopted solution is to equip them with batteries. Others need to be connected to a power outlet or to a USB port. Obviously, batteries allow an easier deployment but they will have to be cyclically replaced. Beacons can work with different communication protocols: iBeacon [23], AltBeacon [24], and Eddystone [25]. The transmitted data, that vary based on the adopted protocol, can be, for example, a Universally Unique IDentifier (UUID), a 24-character string, or a simple URL. Together with the transmitted data, it is also available the RSSI values. Hence, the same considerations

made for Wi-Fi based approaches are valid. Since the information is transmitted via Bluetooth, there is maximum compatibility with smartphones, contrary to what happens with the RFID.

For all the aforementioned reasons, in the AlmaWhere project, we decided to employ BLE beacons. It is worth to notice that, despite the presented advantages, this IoT technology still present open issues that need to be resolved in order to provide a stable, accurate, and scalable localization solution.

2.2. Indoor Navigation Systems for People with Disabilities

Different projects were born with the aim of developing an indoor navigation system to support people with disabilities over the years. They are focused on a specific disability like mobility [26–29] or visual [30,31] or cognitive impairments [32,33]. Analyzing the current literature, it is possible to notice how systems devoted to support visually impaired users are the most studied, employing different approaches like computer vision [34], augmented reality [35,36], RFID [37], and wireless optical communications [38].

Here we provide a description of the more relevant projects that focus on visually impaired users and are based on beacon technology, detailing the chosen approaches, the techniques employed, and the implementation choices made.

NavCog (https://www.cs.cmu.edu/~NavCog/navcog.html Last accessed on 23 February 2020) is probably the most known project devoted to the implementation of a navigation system for visually impaired users [39]. It is based on BLE beacons and exploits the fingerprinting of RSSI values, using Kalman filter and a set of local regression models. This technique requires a distance of about 5–10 m between beacons, that have to be deployed as a grid. A mobile application for iOS is available [31]. The only goal of such an application is to provide turn-by-turn navigation between two points. During navigation, the information can be provided through vocal messages or sonification. The vocal messages speed can be regulated (slow or fast). The map, the POIs (Points of Interest), and the beacons deployment can be updated through a web interface. Another interesting project is StaNavi [40], a navigation system that provides turn-by-turn directions inside the Tokyo Station. It employs BLE beacons too, even if using a different technique: the proximity one. With this technique, the only constraint is to not put beacons densely to avoid conflicts between signals. StaNavi is implemented for the iOS 8 platform, so it runs only on Apple devices. It provides two main functionalities: the Free Roam, which allows getting the current location and detailed information about the surrounding; and the Navigation, thanks to which a user can obtain turn-by-turn directions, from his/her current location to a chosen destination. The path is always computed by the server, that uses Dijkstra's Shortest Path and sends the result in an XML file. The user interacts with the User Interface (UI) via ad hoc gestures. The indications are read using Text-to-Speech APIs, included from iOS 7. Last but not least, there is GuideBeacon [41]. Similar to StaNavi, it localizes the user thanks to the proximity to BLE beacons. To avoid the conflicts caused by the variability of RSSI signals, they developed an algorithm based on a window of the last n signals. Unlike the previous project, the mobile application is developed for the Android OS. It provides only the navigation module, allowing a user to reach a destination from his/her current position. The map of the building is stored on a server. The interaction between the users and the mobile application takes place through Speech-Recognizer and Text-To-Speech.

Finally, even though it regards an outdoor wayfinding system, it is worth mentioning the work of Fogli, Arenghi, and Gentilin [42]. In fact, in the design phase, they exploit a Universal Design approach with the aim of developing a general system able to fit the needs of everyone to find the best path. This was possible thanks to an interdisciplinary team involving not only HCI experts and developers but also civil engineers, architects, and representatives of various communities, including that of people with disabilities.

As stated at the beginning of this Section, the AlmaWhere project presents some similarities, with the above-described projects but it also differs in various aspects. Considering the similarities, it is based on BLE beacons and it employs the proximity technique

to identify the location of users inside the building and it uses the compass to provide accurate turn-by-turn indications. Regarding the differences, we can identify at least four main ones. First of all, it has been designed following an inclusive design approach, aimed to support not only visually impaired users, but also people with disability in general (i.e., mobility impairments) and, more in general, any person interested to explore the indoor environment (e.g., first-time visitors) [43]. AlmaWhere has been developed for both iOS and Android devices. Moreover, AlmaWhere provides two functionalities in addition to the navigation module, the Around You and the Have a Tour ones (as described in the following Sections). Finally, the interaction between users and the mobile application does not use any Speech-Recognizer or Text-To-Speech. This is possible because the UI has been made accessible, in this way visually impaired users can take advantage of the native screen reader of the devices.

3. Design Issues

In this section, we present the main design issues that emerged while designing the pervasive system (i.e., mobile app and infrastructure), considering our specific case study: providing the University of Bologna buildings with an accessible wayfinding system to support students (including students with disabilities), tourists, and visitors.

The system has been designed considering different constraints, such as: (i) the need to use it in buildings with different configurations and layouts, including new buildings and historical ones; (ii) the possibility to scale the system deployment to the whole University of Bologna facilities (covering a critical number of buildings in five cities); (iii) the buildings can become really crowded (impacting the accuracy of the sensors); (iv) information can be also related to historical facts and artifacts, related to the University of Bologna's long history.

3.1. Localization Techniques

Different indoor localization techniques, based on the RSSI, have been proposed in the literature. They can be employed with any technology which provides a signal (e.g., Wi-Fi and BLE beacons). In this Section, we present the four main ones. Each of them has its own strengths and weakness, here evaluated to understand the most suitable technique for our case study.

The most intuitive technique is certainly the proximity [44]. It allows one to locate a device that is within the radius of at least one beacon. The user's (device) position is approximated to one of the beacons. In the case of multiple signals received by the smart device, only the strongest one is considered. Since a user can be located in correspondence of at most one beacon, there is no need to deploy beacons densely, indeed this situation has to be avoided. The main limitation of this technique is surely the location accuracy while the advantage consists in the limited number of beacons required.

Another intuitive technique is lateration, which is a variation of triangulation that employs distances rather than angle measurements [45]. The position of the device is estimated by measuring the distance between the receiver and three or more emitters. The distance can be measured using the RSSI or other measures, such as Time Of Arrival, Time Difference Of Arrival, and Roundtrip Time Of Flight. The combination of two measures usually guarantees more accurate results [44]. This technique assumes that signals are constant over time to be able to accurately locate users reached by at least three signals. Given that beacon signals are very unstable, the performance of this technique easily degrades. Moreover, even if in optimal conditions, to be able to cover every POInt with at least three signals, it is often necessary to set the beacons on high emitting power, causing extensive battery consumption, or to place a higher number of devices.

A more sophisticated technique is fingerprinting [46]. It requires an offline phase and an online one [44]. The offline phase consists of the collection of signals received in each different position of the building. In the online phase, instead, users are localized comparing the signals received by them to the ones collected in the offline phase. Users are located in the position that has the most similar signals to the ones currently received. This approach promises excellent performance thanks to the offline phase performed on-site, but it also suffers from multiple problems [45]. Firstly, it requires the deployment of a dense grid of beacons. Furthermore, as already explained, the beacon signals are not constant over time making the comparison complex. However, the major drawback is that a change in the beacons positioning requires a brand-new off-line phase, which is a time-consuming activity.

Finally, there is Pedestrian Dead Reckoning that can be used only in combination with one of the previous techniques [47]. It is a relative navigation technique that determines the user's position, starting from a known position and then adding successive position displacements estimated through measurements of distance, speed and direction detected (or supposed) by the smartphone's sensors.

A precise comparison among these techniques is hard to carry on. An intuitive approach could be considering the accuracy, in terms of meters, of such algorithms. Each solution proposed in the scientific literature is evaluated with different settings, and using various metrics. Hence, comparing the precision of the different techniques becomes nearly impossible.

In addition, different variables significantly influence the precision of such algorithms, including the physical characteristics of the building, the presence of other signals in the same band, the number of people nearby, and the technical features of the receiving device. All the aforementioned factors lead to very unstable, and consequently unreliable, signals [48]. This can be also observed by the experimental tests conducted with three beacons (same model and same transmission power), deployed in a simple setting, without altering the overall environmental conditions, whose results are reported in Figure 1. As shown, the signals of the beacon1 and beacon2 go up and down, while the signal of the beacon3 has almost the same RSSI.



Figure 1. Beacons' RSSI: beacon1 (red), beacon2 (green), and beacon3 (blue).

In order to alleviate the effects of the unreliability of the beacons' RSSI, some specific measures have been proposed. In particular, such measures aim to eliminate irregularities in measurements employing statistics-based filters, thus using the data they elaborate rather than the actual measurements for calculating the position. Different solutions have been adopted for the lateration technique. Murata et al. [49] employed the particle filter to process the data and combine them with the ones of other sensors integrated into the smartphone. Another variant combines the particle filter with the n-point trilateration [50]. Finally, Chai et al. exploited the Kalman filter [51]. With regard to the fingerprinting technique, Lu et al. combined the k-nearest neighbor with the moving average filter [52] while Sun et al. [53] combined it with a system able to recognize motion patterns previously collected. Instead, the proximity technique is the one that suffers less from the beacons

unreliability. In fact, it is sufficient to arrange and set up the beacons so that their signals do not overlap.

Among all the main localization techniques based on beacon technology and after conducting some tests in different buildings, we decided to employ the proximity technique. Despite it is theoretically the least accurate localization technique, it proved to be sufficiently precise for our case study. Take as an example blind users, as the target audience that, indeed, requires more careful thought considering the provision of turn-by-turn navigation. AlmaWhere is designed to help them on the way to the desired destination inside a building. It is not intended to replace other supports, such as a white cane or a guide dog, which are however necessary to avoid obstacles. Moreover, proximity is the most scalable and maintainable solution. It requires neither a dense grid of beacons nor an off-line phase. In this way, it is easier to map a new building inside the system (i.e., deploy the beacons based on the defined POIs) and/or modify the beacons deployment in any of them.

3.2. Universal and Inclusive Design vs. Design for a Specific Need

The most suitable design process for obtaining an effectively usable software artifact (as well as a usable UI) is still a debate matter by the experts in the field, in particular when users with special needs are involved in the target audience [54]. In [55], the authors referred to the design as "the identification, discussion and resolution of trade-offs", where "trade-offs" are namely situations where it is necessary to renounce to something with the aim of earning something else [42,56]. With specific regards to a UI design, such trade-offs are related to the purposes of the software artifact and of its interface and to the constraints and limitations, that could impose to choose among those purposes. This is why a design problem does not have a unique correct solution or an only right answer, in fact, the solution or answer to a design requirement is strongly dependent on the values, the interests, and the needs of the involved stakeholders and users [57].

Dealing with target users groups that include users with special needs, we have to take into account both accessibility, which could be intended as the goal or as an attribute of the interface, and a proper design, that is a method or the process/path that has to be walked so as to reach the accessibility goal [58]. In literature, two totally different design processes can be applied: a universal and/or inclusive design or a design for a specific need [59]. In the latter, the designers take into account only a specific kind of users and their needs, and the solution to the design requirement meets only such needs, without being suitable and usable for other users with different preferences [60]. From the users' point of view, this could be the best approach, equipping them with the most suitable and tailored UI, designed with the aim of meeting their needs. From the developers' point of view, this could mean the implementation, the management, and the maintenance of different UIs (or of different applications or software artifacts), whenever it is needed to enlarge the target audience [61].

A long discussion is still ongoing related to the former approach, based on the universal and/or inclusive design [62]. Very often, these two terms are considered synonymous and interchangeable. Others proposed definitions for such terms that have unfocused borders, where the differences between them are not so clear [63].

All this said, it is clear that there are no commonly adopted definitions of these terms, nevertheless, we can define universal design as "The design of products and environments to be usable by all people, to the greatest extent possible, without the need for adaptation or specialized design" [64], while inclusive design can be defined as "The design of mainstream products and/or services that are accessible to, and usable by, specific individuals, with specific needs and preferences, and extending this to others, considering the full range of human diversity with respect to ability, language, culture, gender, age, and other forms of human difference" [65]. Another definition of inclusive design is the following one: "inclusive design' is not a fixed set of design criteria, but a constantly evolving philosophy. The goal of creating beautiful and functional environments that can be used equally by everyone, irrespective of age, gender, or disability requires that the design process must be

constantly expanding to accommodate a diverse range of users, as we develop a greater understanding of their requirements, desires and expectations" [63].

In this context, what is clear is that these definitions are based on a central role played by the user and his/her needs, trying to meet as many users as possible, by applying UI adaptation and personalization techniques and approaches. As reported in [59] the role of specific target users in the design phases is becoming strategic: their active involvement in the definition of the design requirement and in all the following phases of a UI design is proved to provide great benefit to the quality of the final result [66]. Thus, intrigued by this issue, we decided to move a further step forward by involving a developer with a disability (a blind person) in the implementation team, so as to experience the so-called "design for user empowerment" [67]. This is based on the idea that some of the best UI and the best interaction systems are obtained when there are people with disabilities on the design and development team, contributing to all aspects of the design and implementation, not just as participants in user studies. This issue can be intended as a deeper form of inclusive design (and development), applying inclusion from also a different point of view [68].

Similar considerations can be done taking into account the development approaches of mobile phone applications, dealing with the two most commonly used operating systems: Android and iOS. Designing and developing native applications is more expensive, time-consuming, and less effective than designing and developing only one mobile application for both of them, but it lets better exploit mobile devices elements, sensors, and functionalities [66]. Choosing just one of the operating system is not an option: on the one side, Android is currently the most commonly used one; while, on the other side, iOS is the operating system used by all the users with visual impairments, thanks to the great support they provide in terms of a dedicated screen reader, quality and precision of the gesture recognition on the multi-touch screen, etc [69]. Thus, the best solution, in this sense, seems to be the use of hybrid applications and related development frameworks.

Summing up, in order to create a software artifact for mobile devices, meeting the needs of a wide target audience (including those users with disabilities), the most suitable and effective design process and app development strategy seem to be the inclusive design and the hybrid app development. Hence, those will be the ones that we will apply in our case study.

4. System Walkthrough

This Section presents the details about how we designed and deployed the pervasive system, addressing the different design issues above mentioned, in terms of: (i) localization strategy and beacons deployment; (ii) the design process; (iii) system architecture and data model. Finally, we present the application implemented, focusing on the main functions that can be enjoyed.

4.1. Localization Module and Beacons Deployment

As already anticipated, we decided to employ proximity as the chosen localization technique. Once implemented it, the right combination between the beacon settings, the parameters of the localization algorithm, and the best beacon deployments must be found. The beacons settings consists in two parameters: Transmission Time and Transmission Power. Low Transmission Time ensures shorter times for localization but implies more power consumption. High Transmission Power allows us to cover a larger area of the building, yet reducing the accuracy of the localization, making overlap likely, and reducing battery life. On the other hand, even if low Transmission Power requires the use of more beacons to map a building, the localization is more precise, the signals overlap are reduced, and the battery life increases.

The proximity algorithm has to detect the user's position while moving inside the building. It is possible that at a given instant, the smartphone receives the signal from two or more beacons. In such cases, it is important to be able to discriminate the most powerful one (i.e., the closest beacon), ignoring at the same time the other signals. The latter situation

could depend not only on overlapping signals but also on spurious readings [70]. To avoid this situation and to make the algorithm more robust, we implemented it using a window of N signals. Within this window, it is considered the closest beacon the one whose signal has been the most powerful at least K times. The choice of N and K is a trade-off between robustness and the position update speed. Larger N and K ensure robustness but imply that the user's position is updated after receiving at least K signals.

Given the chosen localization technique, there are no constraints on the arrangement of beacons. Essentially, the deployment is mainly driven by the location of: (i) the turning points, (ii) the dead-end corridors, and (iii) the POIs. The turning points must be obviously mapped to be able to indicate to the user the correct way in the presence of several possibilities. The dead-end corridors, instead, allow one to reach any place in the building. Finally, the points of interest must be mapped to notify the user when s/he is nearby one of these.

Among the different brands of beacons available, we chose the Smart Beacon SB16-2 produced by Kontakt.io (https://kontakt.io/, Accessed on 23 February 2020). They are Bluetooth 4.2 compliant and, with seven transmission power levels available, have a range of up to 70 m. Thus, they support both the iBeacon and the Eddystone protocols.

The deployment of AlmaWhere took place in three buildings of the University of Bologna, exploiting different contexts and building configurations. To explain the general deployment procedure that we have defined and employed, we describe below the example of the ground floor of the historic Palazzo Riario, headquarters of the Department of History and Cultures, that is depicted in Figure 2.



Figure 2. Maps of the ground floor of the building.

Starting from the map of the floor, we have defined a graph G = (V, E) where V are all the turning points and the ends of the corridors and E represents the set of edges that connect such vertices. As turning points are considered also the stairs and the elevators. The resulting graph of the ground floor is portrayed in the leftmost part of Figure 3 while in the rightmost part is reported the correspondent beacons deployment. This is the minimum quantity of beacons needed to develop a navigation system.

The second part of the procedure consists of associating the POIs to the beacons in the graph. In our system, each POI is associated to one and only one beacon. Instead, a beacon can be associated to zero or N POIs. All the POIs, that are within the range of a beacon, are simply associated to it. However, it is possible that some POIs are not associated to any beacon. In these cases, new beacons have to be introduced so that, in the end, each POI is associated to a beacon. Obviously, the graph changes accordingly, with a new node for each new beacon. The updated graph of the ground floor is reported in the leftmost part of Figure 4. The red points are the nodes added to be able to map some POIs that were

not associated to any beacon. In the rightmost part, instead, there is the corresponding beacons deployment.



Figure 3. Graph of turning POInts and ends of corridors (**leftmost**) and correspondent beacon deployment (**rightmost**).



Figure 4. Complete graph (leftmost) and correspondent beacon deployment (rightmost).

As far as height is concerned, it must be far enough away from the floor to avoid unexpected behavior due to ground reflection. Furthermore, sufficient ground clearance ensures adequate signal propagation even in the presence of many people. Since our testing building (i.e., Palazzo Riario) has very high ceilings, circa 7 m, we decided to place the beacons at a height of about 2.5 m.

Once the beacons have been deployed, we performed several tests to understand the best combination of all the parameters described above. In the end, we decided to employ the Transmission Time of 300 ms, the Transmission Power set to -20 dBm (i.e., a range of 4 m) while N (the window size) and K (the minimum number of signals required within the window to change the closest beacon) were respectively equal to three and two.

4.2. Our Design Process

We focused our design process on the inclusion of target users, driving it into an inclusive dimension. The system has been specifically designed for university communities, with particular attention to students and in particular to students with disabilities. Among them, the system aims to meet the needs of students with visual and/or mobility impairments. This means that it has to be designed so as to bring together two aspects of accessibility: indoor accessibility (in terms of accessibility of the building, by taking into account architectural barriers and facilities) and e-accessibility (in terms of digital accessibility of the system, by taking into account the accessibility of the user interface and of the interaction mechanisms) [71].

Beyond these target groups, other users can benefit from the system, according to the so-called "curb cut effect" [6]: university professors, researchers, and staff (as university communities), tourists interested in the historical buildings within the campus, hosting

university offices and venues, as well as people who visit the campus who do not know the university buildings and areas.

With the aim of reaching the result of an effective and efficient system, we have adopted an inclusive design approach, by actively involving some target users in the design process. In particular, we have involved students with visual impairments. Moreover, a blind computer science student has been involved in the development team, so as to better implement and evaluate an accessible user interface and accessible interaction mechanisms, as well as a better definition of indoor beacon positioning. As already stated in Section 3.2, the involvement of a developer with a disability in the development team gave us the chance of applying the so-called "design for user empowerment" [67], and it has been motivated by the consideration that this is the deepest form of inclusive design and development, since the involvement of persons with disabilities in the development team can significantly contribute in improving all the aspect of the UI and of the interaction flows and mechanisms [68].

Taking into account this latter issue, the involvement of target users in the design phase has resulted in the decision of using the proximity technique: the main reason is that this produces a better balance between the number of beacons needed to equip the building with a wayfinding and navigation system and the fact that people with visual impairments will not give up assistive technologies and devices they use every day while moving independently (such as white canes or guide dogs). Thus, they showed that our system would be used as an additional support [72], without requiring a really accurate positioning and navigation system, but robust enough to use a system that could show some errors in the turning points. Moreover, this design approach has brought also the idea of exploiting assistive technologies already installed in the user's mobile device, together with the most commonly used Text-To-Speech feature. This means that a blind student can interact with our system by means of a mobile app with his/her mobile phone screen reader (keeping his/her preferred configurations and settings).

4.3. System Architecture and Data Model

The system architecture has not undergone major changes compared to previous works [69,73]. The data about the buildings are stored in a database on the server. A web application, implemented using the PHP framework Symfony (https://symfony.com/, Accessed on 23 February 2020), allows us to manipulate them, performing CRUD operations on all the system entities (e.g., buildings, beacons, vertex, edges, ...). Furthermore, the server will expose specific APIs to allow mobile applications to retrieve the data. Once downloaded the data from the server, the mobile application will interact with the beacons to determine the user's position, allowing the user to enjoy the various features. The retrieved data will be used to build a graph and Dijkstra's algorithm will be exploited to find the shortest path between the users' current position and the selected destination. In order to improve the overall scalability of the system, the wayfinding algorithm will be implemented client-side. This solution brings several advantages including less computational load on the server, less bandwidth consumption, and above all a faster recalculation of the path which can significantly affect the user experience.

In Figure 5, we reported the entity-relationship model of AlmaWhere. Each building is characterized by the name, geographical position, and UUID. The building has one or more floors, each of which contains the image of the map that will be shown when the user is on the specific floor. With regard to the navigation graph, there are two main entities: vertexes and edges. Even if the vertexes coincide with the beacons, we kept them in separate tables as the positioning of the beacons inside the building may be subject to changes based on maintenance operations. The beacon contains the UUID, the major, and the minor while no additional information is associated to the vertexes. The edge, instead, connects two vertexes. Even if each edge is bi-directional, it is stored on the database considering one single direction, with the aim of saving space. Together with the origin and the destination nodes, four other pieces of information are saved: the coordinates (x,y) of the map portion

relative to such edge, the degree of difference between the origin and the destination node, which is used to indicate in which direction the user must move (i.e., go ahead/back, turn right/left), and a boolean specifying that such edge is accessible to people with mobility impairments or not. As already stated, zero or N POIs can be associated to a vertex. A POI has the following attributes: name, degree, info, touristic info, and typology. Finally, there are tours, that essentially include ordered lists of POIs.



Figure 5. AlmaWhere Data Model.

4.4. Our System at a Glance

In this section, we detail the functionalities of our mobile application. We implemented it with the React Native framework (https://facebook.github.io/react-native/, Accessed on 23 February 2020), which allowed us to create native apps for Android and iOS using a common Javascript codebase.

4.4.1. Home and Settings

When launched, the app interacts with the server to understand if there is an updated version of the data, and eventually downloads it. Then, it tries to understand the building where the user is located, adopting two different strategies. The first one consists of sensing a beacon and reading its UUID, which is unique for each building. The second one, which is used if the first does not give results, employs GPS to get the current latitude and longitude and compare them with the ones of all buildings. The nearest building is then selected. Anyway, a user can manually select a building.

Once the building has been identified, the homepage, reported on the leftmost part of Figure 6, is shown. It is composed of a title, containing the name of the building, and of five buttons: the first three ones are related to the three main functionalities of the app, the fourth one allows us to change the building, while a tap on the last one open the settings, whose screen is depicted in the rightmost part of Figure 6.

In the Settings view, it is possible to specify whether or not the user has mobility impairments (temporary or not). When checked, this option filters out the edges not marked as accessible in the path computation. Instead, there is no option for visually impaired users, since the UI (User Interface) is accessible (i.e., the screen reader can read the displayed text) and there's no special need in the path computation. The other options allow us to enable the Text-To-Speech, the GPS, and the dark theme. Finally, there are two information panels with a description of the app and the credits.



Figure 6. Homepage (leftmost) and settings (rightmost) of the mobile application.

4.4.2. Navigate to

This functionality takes care of leading the user from his/her current position to a specific destination, chosen from the list of POIs of the building. First of all, the graph is created using the information about the building. Then, the route is computed with Dijkstra's algorithm, using the library *node-dijkstra* (https://github.com/albertorestifo/node-dijkstra, Accessed on 23 February 2020). Once the path is computed, the navigation starts, showing the Navigation UI, reported in Figure 7. It is divided into three sections. The top one contains the next indication to follow. In the center, there is always the portion of the map relative to the edge that the user must travel. Its orientation varies according to the device compass. If there are POIs associated to it, the button "Near" is shown. The bottom bar, instead, contains the buttons "Details" and "Exit".

The indications are provided at the beginning of the navigation and whenever the nearest beacon changes. They are based on the device orientation and on the degree of the edge that the user must travel. In the example of Figure 7, the app has already shown the message to go straight ahead and it now shows the next indication (i.e., turn right) on the top bar and the map relative to the current edge. When the app senses the beacon associated to the turning POInt, the next indication becomes the current one, and it appears over the map for a predefined time interval (currently set to 3 seconds), as shown in the rightmost part of Figure 7. Then, it disappears and the map is updated, together with the next indications in the top bar.

During the navigation, if the app detects a beacon, that is not part of the established route, it recomputes a new route using the last sensed beacon as the starting point. All route computations take place on the device and not on the server to ensure a rapid reaction and full functionality even in the absence of connectivity.

At any time of travel, the user can tap on one of the three buttons. The bottom "Exit" simply interrupts the navigation, showing the homepage. The "Details" one shows a sliding panel containing the list of all the indications to reach the destination, also depicted in the leftmost part of Figure 8. "Near", instead, opens a pop-up with the list of the nearby POIs (i.e., the ones associated to the last sensed beacon). The user can view any cultural information of a POI by clicking on its name, without interrupting the navigation that continues in the background.

Finally, when the destination is reached, a final indication is computed taking into account the orientation of the device and the one of the POI.



Figure 7. Navigation UI: general view of the map (left) and turn signal (right).

14:58 - Next direction: Turn right then keep going straight on	No SIM Next Tui then	15:06 direction: rn right keep going straight on	
e Exit	Close to	o you:	
Turn right	Cortile de	ell'Ercole	
Keep going straight on	Statue of	Vittorio Emanuele	
Keep going straight on			
Arrived			
			Close
			Near
	Details		Exit

Figure 8. Navigation UI: details view (left) and near view (right).

4.4.3. Around You

This feature allows one, once a beacon is sensed, to get the list of the nearby POIs. It has a double aim. On the accessibility side, it allows blind people to build a mind map, becoming familiar with the buildings the first few times s/he visits it. On the museum side, instead, it helps users who see particular POIs in getting more cultural/historical information.

The list of nearby POIs, which are flanked by an icon that indicates whether or not they have a historical value, is reported on the leftmost part of Figure 9 while in the rightmost one, it is depicted the details of a particular POI belonging to the historical Palazzo Poggi (the headquarters of the University of Bologna and of the rector of the university), that is the Pellegrino Rossi statue. The details consist of a photo, a title, and a description, that can be vocalized by the Text-To-Speech.





Figure 9. Around you UI: List of POIs (left) and POI Details (right).

4.4.4. Have a Tour

The last feature of AlmaWhere is "Have a Tour". It does not allow the user to choose a single destination but it guides him/her across a predefined list of POIs that presents some peculiarities from the cultural and artistic Points of view. The idea is to propose a set of theme tours, that allow users to admire some POIs like statues and paintings, present in the building. An example is reported in Figure 10 where the screenshot is reporting the list of POIs included in such a tour.

The indications are provided as in normal navigation, but upon reaching each POI, its descriptive card is shown. When the user has finished reading the description, s/he can continue the tour by tapping the "Continue tour" button.

If there are no tours available in a given building, the button "Have a tour" is not shown on the homepage.



Figure 10. Tour UI.

5. Field Studies

To assess the efficacy of our approach, such as the accessibility and opportunities of use of our system, we conducted three different evaluation sessions in natural settings (i.e., in the University premises). Each field study targeted a specific location (natural setting) and a specific target group, following a purposive sampling (i.e., grouping participants according to preselected criteria). The three evaluations were set up in different scenarios, as briefly presented in Table 1, and described in detail in the next subsections.

Location	Description	Number of Users	Target
Palazzo Riario	The headquarters of the Department of History and Cultures, Bologna. It includes the <i>Tecnolab</i> , a technological laboratory equipped with different assistive technologies, created to welcome students with disabilities.	5	Students with disabilities
Palazzo Poggi	Built in 1549, it is the headquarters of the University of Bologna and of the rector of the university. It also includes a museum, a library, some classrooms, and several administrative offices.	5	Visitors and Tourists
Cesena Campus	Campus Campus It is one of the new Multicampus structures of the University of Bologna build in the city of Cesena. The new campus building hosts different degree courses and department facilities.		Students

Table 1. Details about the three evaluation sessions.

5.1. Methodology

All the sessions were organized following the same protocol. Firstly, we set up the infrastructure, locating the beacons considering the more strategic deployment, following the process described in Section 4.1, on the basis of the building structures and the declared POIs. The experimenters performed some tests to confirm that the infrastructure was set up in the correct way. After, we received the participants. We firstly made an introductory description of the purpose of the research, informing them about what is expected of a research participant, including the amount of time likely to be required for participation and the fact that the participation is voluntary and that one can withdraw at any time with no negative repercussions. Then, we performed the evaluation session, one user at a time. We provided the user with an iPhone 8 or with a Redmi Note 7 (with Android 9) where AlmaWhere was installed, to avoid asking users to install new software on their personal smartphones. We provided users with the possibility to choose Android or iOS to avoid issues due to using an unfamiliar operating system.

Each evaluation session was performed with two researchers, one to assist the user during the wayfinding activity, and the other one, following the session a couple of steps back, collecting data and annotating feedback, reactions, and the precision of the provided step-by-step navigation info. After the session, another researcher performed a structured interview to collect both qualitative and quantitative data, recording the participants for future coding and analysis (after providing them the information on the processing of the collected data and obtaining their oral—considering blind users—or written informed consent, accordingly with the General Data Protection Regulation). The informed consent was approved by the Privacy and Data Protection Office of the University of Bologna, and supervised by the Services for disabled students and students with specific learning disabilities. The structured interview consists of different predefined questions. In particular, the interview comprised different sections: we started with (a) some questions related to general information (age, gender, etc.), we continue with (b) questions related to their familiarity with apps for indoor/outdoor wayfinding, then, (c) specific questions for each function (i.e., "Navigate to" and "Around you") were asked, using a five point Likert scale (from "very satisfied" to "very dissatisfied", including the neutral value), finally, we conclude the interview with (d) an open question to collect general feedback and comments. Each participant went through all the sections, but the questions were slightly different in accordance with the different targets considered, to gather relevant aspects peculiar to each group's needs.

5.1.1. Evaluation with Students with Disabilities

We organized a session with five students with disabilities (four blind users and one wheelchair user) in the main building of the Department of History and Cultures. Students (two females) were from 19 to 44 years old (mean = 26.4 and sd = 8.8), enrolled in different courses, such as Computer Science, Law, and Arts, Humanities, and Cultural Heritage. We selected such a building since it includes the Tecnolab, a technological laboratory created to welcome students with disabilities, equipping them with different assistive technologies they can try and explore within the lab, together with specialized tutors. The lab has been created in collaboration with the Services for disabled students and students with specific learning disabilities of the University of Bologna (in the Italian language: "ABIS—Settore Diritto allo studio—Ufficio Servizi per gli studenti con disabilità e con DSA"). We asked participants to: 1. start from the main entrance and to look for the closer toilet (for females/males/wheelchair users); 2. once at the toilet, check for nearby POIs; 3. from the toilet, reach the library at the 1 floor (using the accessible elevator or the stairs, on the basis of the user's needs). All the students were familiar only with the Tecnolab premises and not with the building itself.

5.1.2. Evaluation with Visitors and Tourists

We randomly engaged five users who voluntarily accepted to participate in the study: we simply approached them close to the main entrance/exit of Palazzo Poggi. In particular, we involved two tourists from Brazil (mother, 54 years old and son, 19 years old) who were there to visit the Poggi Museum, and three students, from 20 to 24 years old (two females) there to visit, for the first time, the library and the famous classroom dedicated to Giosué Carducci, in which the poet gave lectures on the Italian language and literature for 40 years.

In this case, we asked users to: 1. reach the Pellegrino Rossi statue, then, 2. retrieve information about the statue, and, finally, 3. perform a short tour (selecting the only one available on the prototype).

5.1.3. Evaluation with Campus Students

We involved eight students (one female) attending classes at the Campus of Cesena. These students were from 23- to 26-year-old (mean = 23.6, sd = 1.2; one female). One of the participants was experiencing a temporary mobility impairment (broken leg). In this evaluation, we only asked users to reach a particular not well-known classroom inside the Campus.

5.2. Results and Discussion

Figures 11 and 12 show the outputs of the questions related to the "Navigate To" and "Around you" functions, considering five dimensions: Usability, Efficiency, Clarity, Useful in familiar places, Useful in unknown places. The outcome showed that, on average, the users found both the functions usable, effectiveness to use, and clear, with values close to 5 (the maximum). Instead, negative values (less than 3) were obtained when we asked about the usefulness of the function in familiar places. As expected, values were particularly

negative when considering the "Navigate To" function (less than 2). Values were a bit higher considering the "Around You" function in familiar places, at least considering students with disabilities and tourists and visitors. That can be explained considering the fact that using such a function, it is possible to learn information about points of interest. A point of interest could be a historical artifact (considering visitors and tourists), or a toilet (accessible or not), for females or males (considering users with disabilities), or vending machines for coffee and drinks (considering users with visual impairments). Nonetheless, once one became familiar with a place, this function also lost interest. Accordingly, students with visual impairments mostly picked the neutral value (3) to rank this function. The reason is that they loved the function and they think it is incredibly useful in unknown places, but, once one is getting used to a place, s/he should not need it anymore. Focusing on visitors, one commented: "I think the function can be really interesting the first time you use it (independently if the place is familiar or not). Then, the statue, for example, will be always the same. This is why I ranked 1". Contrariwise, a visitor who assigned four to the function, claimed: "Also if you come here [Palazzo Poggi] often, I don't think this statue will attract your attention, so you need a system that can facilitate this and let people learn about the story of the represented person". Considering the results gathered in the Cesena Campus, almost all students assigned one to the "Around You" function in familiar places. Checking the data, it is possible to observe that they assigned a low score also when using such a function in unknown places. The motivation can be found considering that the Cesena Campus is a completely new building, with no artwork or historical artifacts. Nonetheless, we are planning to make the function more engaging, including information about the campus, such as lesson schedules, professors' reception hours. A very encouraging result is that, in all the field studies, the "Navigation To" function obtained the maximum score (5) when considering unknown places.

Focusing on the users with disabilities, different insights emerged. On the one hand, users with mobility impairments really appreciated the system, being surprised by the calculated path including the accessibility facilities (e.g., accessible elevators), avoiding barriers (e.g., stairs). On the other hand, three out of four blind users were expecting a more precise system in terms of step-by-step navigation, but, despite that, they recognized the relevance of such a system: none of the participants ever used a similar system for indoor navigation, and they were positively impressed. One limitation that emerged during the study was related to the smartphone orientation. In fact, the application exploits also the compass to provide information to the user, but visually impaired people often prefer to keep the phone in a bag (using headphones to control it) or keep it in a landscape position, to have the micro close to the ear. To solve this issue, we decided to allow users to disable the compass and rely only on the beacons mapping.



Figure 11. "Navigate To" function: Likert scale average values, comparing the different target users.

Usability

Efficiency



Useful in

Useful in

familiar places unknown places

Figure 12. "Around you" function: Likert scale average values, comparing the different target users.

Clarity

Tourists and visitors really liked the "Around You" and the "Tour" functions. One tourist said: "The Around you function is super interesting and, in this building [Palazzo Poggi] full of historical facts, it can become a very important tool, a kind of virtual guide, that allow you to explore every corner of this place". At the same way, they all appreciate the "Tour" function. One tourist claimed: "When I enter this building, I felt very confused. It is huge and very complicated to navigate. Having an app that can provide you step-bystep navigation following touristic tours can really enhance the visitor's experience of such a beautiful and historical place".

Students in Cesena understand the idea behind the application and, in general, appreciate it. Nonetheless, most of them suggested to alleviate the loss of interest in using the app in familiar places by including some other information related to the University premises and the user's position, such as the class schedule. Accordingly, one reported: "The app is interesting and can provide relevant information to reach a Professor's office or a lab for the first time, but then, it has to provide me something more, something more dynamic, such as the lessons' timetable, otherwise I stop using it". Moreover, as already anticipated, students (without disabilities) in the new Cesena campus also didn't completely understand the potential of the "Around you" function, due to the absence of historical artifacts or facts and the possibility to visualize the point of interests while facing them. This is an interesting result that can be taken into account in refining the app. In fact, the app can simply allow each user to personalize the app, choosing which functions to enable and when.

Due to the limited number of participants who tested the system, we can not provide the statistical significance of the collected quantitative data. Nonetheless, we can state that the general feeling was very positive and participants enjoyed using it.

6. Conclusions and Future Work

In this paper, we present how a mobile application, together with an IoT infrastructure (exploiting BLE beacons) can be exploited to locate a user inside an indoor environment and to provide: (i). wayfinding real-time instructions; ii. location-based information about the mapped points of interest; (ii). predefined tours, including different points of interest. The system has been designed to be accessible, following an inclusive design approach, in order to address the needs and preferences of different target users, including blind users, mobility-impaired users, and visitors/tourists. Three evaluation tests have been carried out to assess the enjoyability and accuracy of the system, showing that the system has a good trade-off between the facility of use and the accuracy of the user's positioning. Moreover, thanks to its low-cost and easy deployment process, it can be scalable and highly adaptable to different building settings and layouts, addressing our initial constraints.

Reflecting on the lessons learned during the design and implementation of our case study, we can confirm that following an inclusive design approach was strategic in achieving positive results, in terms of usability, efficiency, and clarity of the system, for all the users involved, confirming previous findings [56]. In fact, we defined the requirements considering different target groups (users with disability, tourists and visitors, and students), with specific needs and preferences, and we design a system able to satisfy the different needs, keeping accessibility in mind and taking advantage of the curb-cut effect [6]. Moreover, having the possibility to put into practice the "design for user empowerment" approach, having a blind user in the development team, prevent our system to include issues related to the accessibility and usability of the system, especially for low vision and blind people [67].

As future work, we just started to investigate the use of conversational user interfaces (such as the Amazon Alexa) in combination with our system, to understand if it can be useful in assisting users during wayfinding tasks.

Author Contributions: Conceptualization, S.M. and P.S.; methodology, C.P.; software, G.D.; writing original draft preparation, G.D. and C.P.; writing—review and editing, S.M. and P.S.; supervision, S.M.; project administration, S.M.; funding acquisition, P.S. and S.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: No applicable.

Acknowledgments: The authors want to thank Rabih Chattat, Constantino Marmo, Donatella Alvisi, Vincenzo Tamburelli, and Sonia Duca: this project was born thanks to them, their availability, and their involvement. A special thank goes to the University of Bologna Services for disabled students and students with specific learning disabilities and to all the students involved in the evaluation phases. Finally, the authors thank Francesco Vignola, Elena Arlotti, Enrico Fantini, Simone Massacesi, Stefano Valentini, Andrea Zanuccoli, Leonardo Papini, and Lorenzo Monti for their precious support during the design and the development of the system, and Vincenzo Rubano and Giacomo Mambelli, who implemented and refined the system.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

- PoI Points of Interest
- BLE Bluetooth Low Energy
- GPS Global Positioning System
- RSSI Received Signal Strength Indicator
- RFID Radio-Frequency IDentification
- IoT Internet of Things
- UUID Universally Unique IDentifier
- UI User Interface

References

- Mirri, S.; Prandi, C.; Salomoni, P. A context-aware system for personalized and accessible pedestrian paths. In Proceedings of the 2014 International Conference on High Performance Computing & Simulation (HPCS), Bologna, Italy, 21–25 July 2014; pp. 833–840.
- Bellini, G.; Cipriano, M.; Comai, S.; De Angeli, N.; Gargano, J.P.; Gianella, M.; Goi, G.; Ingrao, G.; Masciadri, A.; Rossi, G.; et al. Understanding Social Behaviour in a Health-Care Facility from Localization Data: A Case Study. *Sensors* 2021, 21, 2147. [PubMed]
- 3. Kunhoth, J.; Karkar, A.; Al-Maadeed, S.; Al-Ali, A. Indoor positioning and wayfinding systems: a survey. *Hum.-Centric Comput. Inf. Sci.* **2020**, *10*, 1–41. [CrossRef]
- Kim, J.E.; Bessho, M.; Koshizuka, N.; Sakamura, K. Enhancing public transit accessibility for the visually impaired using IoT and open data infrastructures. In Proceedings of the First International Conference on IoT in Urban Space, Rome, Italy, 27–28 October 2014; pp. 80–86.

- 5. Prandi, C.; Monti, L.; Ceccarini, C.; Salomoni, P. Smart campus: Fostering the community awareness through an intelligent environment. *Mob. Netw. Appl.* **2020**, *25*, 945–952.
- 6. Treviranus, J. Leveraging the web as a platform for economic inclusion. Behav. Sci. Law 2014, 32, 94–103.
- Hofmann-Wellenhof, B.; Lichtenegger, H.; Collins, J. Global Positioning System: Theory and Practice; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2012.
- 8. Tian, Y.; Yang, X.; Yi, C.; Arditi, A. Toward a computer vision-based wayfinding aid for blind persons to access unfamiliar indoor environments. *Mach. Vis. Appl.* **2013**, *24*, 521–535. [CrossRef]
- 9. Lin, B.S.; Lee, C.C.; Chiang, P.Y. Simple smartphone-based guiding system for visually impaired people. *Sensors* **2017**, *17*, 1371. [CrossRef] [PubMed]
- 10. Manduchi, R.; Coughlan, J. (Computer) Vision Without Sight. Commun. ACM 2012, 55, 96–104. [CrossRef]
- 11. Tapu, R.; Mocanu, B.; Zaharia, T. A computer vision system that ensure the autonomous navigation of blind people. In Proceedings of the 2013 E-Health and Bioengineering Conference (EHB), Iasi, Romania, 21–23 November 2013.
- Kuo, Y.S.; Pannuto, P.; Hsiao, K.J.; Dutta, P. Luxapose: indoor positioning with mobile phones and visible light. In Proceedings of the MobiCom '14 Proceedings of the 20th Annual International Conference on Mobile Computing and Networking, Maui, HI, USA, 7–11 September 2014; pp. 447–458.
- 13. Nakajima, M.; Haruyama, S. New indoor navigation system for visually impaired people using visible light communication. *EURASIP J. Wirel. Commun. Netw.* **2013**, 2013, 37. [CrossRef]
- 14. Singh, V.; Aggarwal, G.; Ujwal, B. Ensemble based real-time indoor localization using stray WiFi signal. In Proceedings of the 2018 IEEE International Conference on Consumer Electronics (ICCE), Las Vegas, NV, USA, 12–14 January 2018; pp. 1–5.
- 15. Chen, Z.; Zou, H.; Yang, J.; Jiang, H.; Xie, L. WiFi fingerprinting indoor localization using local feature-based deep LSTM. *IEEE Syst. J.* **2019**, *14*, 3001–3010. [CrossRef]
- Gallagher, T.; Wise, E.; Li, B.; Dempster, A.G.; Rizos, C.; Ramsey-Stewart, E. Indoor positioning system based on sensor fusion for the Blind and Visually Impaired. In Proceedings of the 2012 International Conference on Indoor Positioning and Indoor Navigation, Sydney, NSW, Australia, 13–15 November 2012.
- 17. Roussos, G.; Kostakos, V. RFID in pervasive computing: state-of-the-art and outlook. *Pervasive Mob. Comput.* **2009**, *5*, 110–131. [CrossRef]
- Bianchi, F.; Masciadri, A.; Salice, F. ODINS: On-Demand Indoor Navigation System RFID Based. *Stud. Health Technol. Inform.* 2015, 217, 341–349.
- Biader Ceipidor, U.; D'Atri, E.; Medaglia, C.M.; Mei, M.; Serbanati, A.; Azzalin, G.; Rizzo, F.; Sironi, M.; Contenti, M.; D'Atri, A. A RFID System to Help Visually Impaired People in Mobility. In Proceedings of the EU RFID Forum 2007, Brussels, Belgium, 13–14 March 2007.
- 20. Faria, J.; Lopes, S.; Fernandes, H.; Martins, P.; Barroso, J. Electronic white cane for blind people navigation assistance. In Proceedings of the 2010 World Automation Congress, Kobe, Japan, 19–23 September 2010.
- Chumkamon, S.; Tuvaphanthaphiphat, P.; Keeratiwintakorn, P. A Blind Navigation System Using RFID for Indoor Environments. In Proceedings of the 2008 5th International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technolog, Krabi, Thailand, 14–17 May 2008.
- 22. Jeon, K.E.; She, J.; Soonsawad, P.; Ng, P.C. BLE beacons for Internet of Things applications: Survey, challenges, and opportunities. *IEEE Internet Things J.* 2018, *5*, 811–828. [CrossRef]
- 23. Newman, N. Apple iBeacon technology briefing. J. Direct Data Digit. Mark. Pract. 2014, 15, 222–225. [CrossRef]
- 24. Helms, D. *Altbeacon Protocol Specification*, v1.0. 2017. Available online: https://github.com/AltBeacon/spec (accessed on 28 April 2021).
- Amadeo, R. Meet Google's 'Eddystone'-a Flexible Open Source iBeacon Fighter. Ars Technica 2015. Available online: https: //arstechnica.com/gadgets/2015/07/meet-googles-eddystone-a-flexible-open-source-ibeacon-fighter/ (accessed on 28 April 2021).
- 26. Oliveira, L.C.d.; Soares, A.B.; Cardoso, A.; Andrade, A.d.O.; Lamounier Júnior, E.A. Mobile Augmented Reality enhances indoor navigation for wheelchair users. *Res. Biomed. Eng.* 2016, *32*, 111–122. [CrossRef]
- 27. Abascal, J.; Lafuente, A.; Marco, A.; Falcó, J.M.; Casas, R.; Sevillano, J.L.; Cascado, D.; Lujan, C. An architecture for assisted navigation in intelligent environments. *Int. J. Commun. Networks Distrib. Syst.* **2010**, *4*, 49–69.
- Amaral, P.F.; García, J.C.G.; Bastos Filho, T.F.; Mazo, M. Ambient assisted route planner based on xml files with accessibility information. In Proceedings of the 2009 IEEE International Symposium on Intelligent Signal Processing, Budapest, Hungary, 26–28 August 2009; pp. 147–152.
- 29. Pereira, C.; Sousa, A.; Filipe, V. Open-Source indoor navigation system adapted to users with motor disabilities. *Procedia Comput. Sci.* **2015**, *67*, 38–47. [CrossRef]
- Watanabe, C.; Minagawa, J. Event Venue Navigation for Visually Impaired People. In Proceedings of the 2019 IEEE International Conference on Big Data and Smart Computing (BigComp), Kyoto, Japan, 27 Februay–2 March 2019; pp. 1–6.
- Ahmetovic, D.; Gleason, C.; Ruan, C.; Kitani, K.; Takagi, H.; Asakawa, C. NavCog: a navigational cognitive assistant for the blind. In Proceedings of the MobileHCI '16 18th International Conference on Human-Computer Interaction with Mobile Devices and Services, Florence, Italy, 6–9 September 2016; pp. 90–99.

- 32. Chang, Y.J.; Wang, T.Y. Indoor wayfinding based on wireless sensor networks for individuals with multiple special needs. *Cybern. Syst. Int. J.* **2010**, *41*, 317–333. [CrossRef]
- García-Catalá, M.; Rodríguez-Sánchez, M.; Martín-Barroso, E. Survey of indoor location technologies and wayfinding systems for users with cognitive disabilities in emergencies. In *Behaviour & Information Technology*; Informa UK Limited: London, UK, 2020; pp. 1–25.
- Taylor, B.; Lee, D.J.; Zhang, D.; Xiong, G. Smart phone-based Indoor guidance system for the visually impaired. In Proceedings of the 2012 12th International Conference on Control Automation Robotics & Vision (ICARCV), Guangzhou, China, 5–7 December 2012; pp. 871–876.
- 35. Al-Khalifa, S.; Al-Razgan, M. Ebsar: Indoor guidance for the visually impaired. Comput. Electr. Eng. 2016, 54, 26–39. [CrossRef]
- Yu, X.; Yang, G.; Jones, S.; Saniie, J. AR Marker Aided Obstacle Localization System for Assisting Visually Impaired. In Proceedings of the 2018 IEEE International Conference on Electro/Information Technology (EIT), Rochester, MI, USA, 3–5 May 2018; pp. 0271–0276.
- Lim, K.L.; Seng, K.P.; Yeong, L.S.; Ang, L.M. RFID and dead-reckoning-based indoor navigation for visually impaired pedestrians. In Smart Technologies: Breakthroughs in Research and Practice; IGI Global: Hershey, PA, USA; 2018; pp. 1–16.
- Perez-Jimenez, R.; Rabadan, J.; Guerra, V.; Aguiar, L.; Rufo, J. Fundamentals of Indoor VLP: Providing Autonomous Mobility for Visually Impaired People. In Proceedings of the 2017 International Conference and Workshop on Bioinspired Intelligence (IWOBI), Funchal, Portugal, 10–12 July 2017; pp. 1–6.
- 39. Murata, M.; Ahmetovic, D.; Sato, D.; Takagi, H.; Kitani, K.M.; Asakawa, C. Smartphone-based localization for blind navigation in building-scale indoor environments. *Pervasive Mob. Comput.* **2019**, *57*, 14–32. [CrossRef]
- Kim, J.E.; Bessho, M.; Kobayashi, S.; Koshizuka, N.; Sakamura, K. Navigating visually impaired travelers in a large train station using smartphone and bluetooth low energy. In Proceedings of the SAC'16 Proceedings of the 31st Annual ACM Symposium on Applied Computing, Fortaleza, Ceara, Brazil, 16–20 March 2016; pp. 604–616.
- Cheraghi, S.A.; Namboodiri, V.; Walker, L. GuideBeacon: Beacon-based indoor wayfinding for the blind, visually impaired, and disoriented. In Proceedings of the 2017 IEEE International Conference on Pervasive Computing and Communications (PerCom), Kona, HI, USA, 13–17 March 2017.
- 42. Fogli, D.; Arenghi, A.; Gentilin, F. A universal design approach to wayfinding and navigation. In *Multimedia Tools and Applications*; Springer: Berlin/Heidelberg, Germany, 2019; pp. 1–25.
- 43. Clarkson, P.J.; Coleman, R.; Keates, S.; Lebbon, C. Inclusive Design: Design for the Whole Population; Springer: London, UK; 2013.
- 44. Liu, H.; Darabi, H.; Banerjee, P.; Liu, J. Survey of wireless indoor positioning techniques and systems. *IEEE Trans. Syst. Man Cybern.*—*Part C Appl. Rev.* 2007, 37, 1067–1080. [CrossRef]
- 45. Seco, F.; Jimenez, A.R.; Prieto, C.; Roa, J.; Koutsou, K. A survey of mathematical methods for indoor localization. In Proceedings of the 2009 IEEE International Symposium on Intelligent Signal Processing, Budapest, Hungary, 26–28 August 2009.
- 46. Jing, H.; Pinchin, J.; Hill, C.; Moore, T. Wi-Fi fingerprinting based on collaborative confidence level training. *Pervasive Mobile Comput.* **2016**, *30*, 32–44. [CrossRef]
- 47. Beauregard, S.; Haas, H. Pedestrian Dead Reckoning: A Basis for Personal Positioning. In Proceedings of the 3rd Workshop on Positioning, Navigation and Communication, Hannover, Germany, 16 March 2006.
- Dong, Q.; Dargie, W. Evaluation of the reliability of RSSI for indoor localization. In Proceedings of the 2012 International Conference on Wireless Communications in Underground and Confined Areas, Clermont-Ferrand, France, 28–30 August 2012; pp. 1–6.
- Murata, M.; Ahmetovic, D.; Sato, D.; Takagi, H.; Kitani, K.M.; Asakawa, C. Smartphone-based indoor localization for blind navigation across building complexes. In Proceedings of the 2018 IEEE International Conference on Pervasive Computing and Communications (PerCom), Athens, Greece, 19–23 March 2018; pp. 1–10.
- 50. Zafari, F.; Papapanagiotou, I. Enhancing ibeacon based micro-location with particle filtering. In Proceedings of the 2015 IEEE Global Communications Conference (GLOBECOM), San Diego CA, USA, 1–7 December 2015; pp. 1–7.
- 51. Chai, S.; An, R.; Du, Z. An indoor positioning algorithm using bluetooth low energy RSSI. In Proceedings of the 2016 International Conference on Advanced Materials Science and Environmental Engineering, Chiang Mai, Thailand, 26–27 June 2016.
- 52. Lu, T.T.; Yeh, S.C.; Chen, C.Y. A study of indoor positioning systems using iBeacons with different transmission power levels. *J. Chin. Inst. Eng.* **2017**, *40*, 525–535. [CrossRef]
- Sun, W.; Liu, J.; Wu, C.; Yang, Z.; Zhang, X.; Liu, Y. MoLoc: On distinguishing fingerprint twins. In Proceedings of the 2013 IEEE 33rd International Conference on Distributed Computing Systems, Philadelphia, PA, USA, 8–11 July 2013; pp. 226–235.
- 54. Fischer, G. Identifying and exploring design trade-offs in human-centered design. In Proceedings of the 2018 International Conference on Advanced Visual Interfaces, Castiglione della Pescaia, Italy, 29 May–1 June 2018; pp. 1–9.
- 55. Rosson, M.B.; Carroll, J.M.; Hill, N. Usability Engineering: Scenario-Based Development of Human-Computer Interaction; Morgan Kaufmann: Elsevier: Burlington, MA, USA, 2001.
- 56. Fogli, D.; Arenghi, A. '*Design for All'versus' One-Size-Fits-All': The Case of Cultural Heritage*. In Proceedings of the 5th Workshop on Cultures of Participation in the Digital Age: Design Trade-offs for an Inclusive Society, Grosseto, Italy, 29 May–1 June 2018.
- 57. Fischer, G. Exploring design trade-offs for quality of life in human-centered design. Interactions 2017, 25, 26–33. [CrossRef]
- 58. Holmes, K. *Mismatch: How Inclusion Shapes Design;* MIT Press: Cambridge, MA, USA, 2018.
- 59. Story, M.F. Principles of universal design. In Universal Design Handbook; McGraw-Hill Professional: New York, NY, USA, 2001.

- 60. Ayotte, D.; Vass, J.; Mitchell, J.; Treviranus, J. Personalizing interfaces using an inclusive design approach. In Proceedings of the International Conference on Universal Access in Human-Computer Interaction, Heraklion, Greece, 22–27 June 2014; pp. 191–202.
- 61. Mitchell, J.; Treviranus, J. Inclusive Design in Ecosystems. In *E-Health Two-Sided Markets*; Elsevier: Amsterdam, The Netherlands, 2017; pp. 43–61.
- 62. Newell, A. Inclusive design or assistive technology. In Inclusive Design; Springer: Berlin/Heidelberg, Germany, 2003; pp. 172–181.
- Persson, H.; Åhman, H.; Yngling, A.A.; Gulliksen, J. Universal design, inclusive design, accessible design, design for all: Different concepts—One goal? On the concept of accessibility—Historical, methodological and philosophical aspects. *Univers. Access Inf. Soc.* 2015, *14*, 505–526. [CrossRef]
- 64. Story, M.F. Maximizing usability: The principles of universal design. Assist. Technol. 1998, 10, 4–12. [CrossRef] [PubMed]
- 65. Treviranus, J. The Value of Being Different. In Proceedings of the 16th Web for All 2019 Personalization-Personalizing the Web, Francisco, CA, USA, 13–14 May 2019; pp. 1–7.
- 66. Mirri, S.; Roccetti, M.; Salomoni, P. Collaborative design of software applications: the role of users. *Hum.-Centric Comput. Inf. Sci.* **2018**, *8*, 6. [CrossRef]
- 67. Ladner, R.E. Design for user empowerment. *Interactions* **2015**, *22*, 24–29. [CrossRef]
- 68. Brewer, J. Exploring paths to a more accessible digital future. In Proceedings of the 20th International ACM SIGACCESS Conference on Computers and Accessibility, Galway, Ireland, 22–24 October 2018; pp. 1–2.
- Delnevo, G.; Monti, L.; Vignola, F.; Salomoni, P.; Mirri, S. AlmaWhere: A prototype of accessible indoor wayfinding and navigation system. In Proceedings of the 2018 15th IEEE Annual Consumer Communications & Networking Conference (CCNC), Las Vegas, NV, USA, 12–15 January 2018; pp. 1–6.
- Faragher, R.; Harle, R. An analysis of the accuracy of bluetooth low energy for indoor positioning applications. In Proceedings of the 27th International Technical Meeting of The Satellite Division of the Institute of Navigation (ION GNSS+ 2014), Tampa, Florida, USA, 8–12 September 2014; Volume 812, pp. 201–210.
- Mirri, S.; Prandi, C.; Salomoni, P.; Callegati, F.; Melis, A.; Prandini, M. A service-oriented approach to crowdsensing for accessible smart mobility scenarios. *Mob. Inf. Syst.* 2016, 2016, 2821680. [CrossRef]
- Mirri, S.; Prandi, C.; Salomoni, P. Personalizing Pedestrian Accessible way-finding with mPASS. In Proceedings of the 2016 13th IEEE Annual Consumer Communications & Networking Conference (CCNC), Las Vegas, NV, USA, 9–12 January 2016; pp. 1119–1124.
- Delnevo, G.; Mambelli, G.; Rubano, V.; Prandi, C.; Mirri, S. Almawhere 2.0: a pervasive system to facilitate indoor wayfinding. In Proceedings of the 2020 17th IEEE Annual Consumer Communications & Networking Conference (CCNC), Las Vegas, NV, USA, 10–13 January 2020.